Author’s response to reviews

Title: Realising the right to sexual and reproductive health: Access to essential medicines for medical abortion as a core obligation

Authors:
Katrina Perehudoff (katrina.perehudoff@gmail.com; s.j.k.perehudoff@umcg.nl)
Lucía Berro Pizzarossa (lberropizzarossa@gmail.com)
Jelle Stekelenburg (jelle.stekelenburg@online.nl)

Version: 1 Date: 03 Oct 2017

Author’s response to reviews:

October 3, 2017

Dear Mrs. Pafitis,

Thank you for your thoughtful remarks on our manuscript. We have considered your recommendations and those from the reviewers. We are pleased to resubmit the manuscript with the following principle changes:

Minor adaptations to focus the introductory paragraph on the negative health impact of unsafe abortion and on WHO’s role as the only UN agency to endorse the evidence for safe abortion;
Inclusion of CEDAW as an important source of international human rights law related to women’s right to health.

We have responded to the reviewers’ comments in full the table in attachment and amended our manuscript accordingly.
Please find two attachments to this message (responses to reviewers and revised manuscript).

Thank you for your consideration.

Kind regards,

Katrina Perehudoff